Author's response to reviews

Title: Screening and brief interventions for hazardous and harmful alcohol use in primary care: a cluster randomised controlled trial protocol

Authors:

Eileen Kaner (e.f.s.kaner@newcastle.ac.uk)
Martin Bland (mb55@york.ac.uk)
Paul Cassidy (paul.cass@blueyonder.co.uk)
Simon Coulton (s.coulton@kent.ac.uk)
Paolo DeLuca (paolo.deluca@kcl.ac.uk)
Colin Drummond (colin.drummond@kcl.ac.uk)
Eilish Gilvarry (eilish.gilvarry@ntw.nhs.uk)
Christine Godfrey (cg2@york.ac.uk)
Nick Heather (nick.heather@unn.ac.uk)
Judy Myles (jmyles@sgul.ac.uk)
Dorothy Newbury-Birch (dorothy.newbury-birch@newcastle.ac.uk)
Adenekan Oyefeso (a.oyefeso@sgul.ac.uk)
Steve Parrott (sjp22@york.ac.uk)
Katherine Perryman (katherine.perryman@iop.kcl.ac.uk)
Tom Phillips (tom.phillips@humber.nhs.uk)
Don Shenker (dshenker@alcoholconcern.org.uk)
Jonathan Shepherd (j.shepherd@cardiff.ac.uk)

Version: 2  Date: 13 July 2009

Author's response to reviews:

Dear Editor, I have amended the abstract as requested and added the details of the trial registration at the end of the abstract.